Author’s response to reviews

Title: Quantitative analysis of left atrial function in asymptomatic patients with beta-thalassemia major using real-time three-dimensional echocardiography

Authors:

Constantina Aggeli (dina.aggeli@gmail.com)
Ioannis Felekos (yannisfd@hotmail.com)
Emmanouil Poulidakis (povlidakis@yahoo.com)
Athanasios Aggelis (agelal@otenet.gr)
Dimitrios Tousoulis (drtousoulis@hotmail.com)
Christodoulos Stefanadis (cstefan@med.uoa.gr)

Version: 3 Date: 10 November 2011

Author’s response to reviews: see over
Dear Production team

Upon receiving your e-mail we proceeded to the necessary changes.

Formatting changes to be made:

----------------------------------

Major revisions (we require the author to make these changes)

Ethical approval: Any experimental research that is reported in the manuscript should have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Authors’ reply: We have done so in the methods section (study population paragraph).

Figures: 'Graph 1' should be uploaded as a figure file and numbered as such. The legend should be added to the figure legends section and the figure should be cited in numerical order with the other figures.

Authors’ reply: Done

Additional files: All additional files must be mentioned in the text in numerical order, or removed from the system.

Authors’ reply: Done

Additional files: Please ensure all files are referred to as additional files (not, for example, supplementary data). E.g. 'Additional file 1'

Authors’ reply: Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Email address: Please remove the emails from the author list and include email addresses for all the authors on the title page, using the following format:

AB (author initials): abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Authors' reply: Done

Highlighting/tracking: Please remove all highlighting and tracking (underlining) from the manuscript. Please ensure all text is black.

Authors' reply: Done

Abbreviations: Please format these in a paragraph.

Authors' reply: Done

Structure: Please move the figure legends immediately after the reference list.

Authors' reply: Done

Please move the additional materials to the end of the manuscript (after the tables) and re-title this 'Additional files'.

Authors' reply: Done

Tables: Please remove the visible vertical lines from your tables. Please also move the table title to above the table and the legend to below the table, within the text.

Authors' reply: Done

Figures/Tables: Please do not use ranges when listing tables and figures. Please list all the tables/figures, for example: Figures 5, 6, 7, 8 and 9.

Authors' reply: Done
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Authors’ reply: Done

Thank you

-----------------------------

Please return your revisions by 16/11/2011.

We do hope that we have fulfilled your requirements. Please do not hesitate to contact us if further modifications are needed.

Best regards,

Constantina Aggeli, MD

Ioannis Felekos, MD